Author's response to reviews

Title: Androgen secreting steroid cell tumor of the ovary in a young lactating women with acute onset severe hyperandrogenism

Authors:

Altarf Gauhar Dr (altafhaji786@yahoo.com)
Shekhar Sharma Dr (saad_chao123@yahoo.com)
Vijay DK Kumar Dr (sclwm@yahoo.com)

Version: 2 Date: 18 July 2007

Author's response to reviews:

As per the instructions following things were included-
1. Title page
2. Conclusion
3. Pt. Consent
4. Interest
5. Funding